Author’s response to reviews

Title: Association of 4p14 and 6q27 Variation with Graves Disease: a Case-control Study and a Meta-analysis of Available Evidence

Authors:
Xun Chu (chux@chgc.sh.cn)
Fa-Mei Li (lifamei2007@126.com)
Lin LIU (liulin_sd@sina.com)
Li-Nan Pang (690891659@qq.com)
Min SHEN (shenm@chgc.sh.cn)
Hong-Wen LU (13563612986@163.com)
Xiao-Hong ZHANG (zhangxh71129@163.com)

Version: 1 Date: 06 Aug 2016

Author’s response to reviews:

August 6, 2016

Dear Dr. Matteo Pasini,

Thank you very much for your kind consideration of our manuscript (MGTC-D-16-00094) and your valuable comments. We have revised our manuscript according to your comments. Sufficient detail was provided in the methods section as required by the PRISMA guidelines for reporting of meta-analysis studies.

We hope that we have adequately addressed your concerns. Thank you very much for your consideration of this manuscript.

Sincerely,

Xun CHU, PhD
Associated Professor, the State Key Laboratory of Medical Genomics, Chinese National Human Genome Center, Shanghai, China
Tel: 86 21 50801795
Fax: 86 21 51714506
E-mail: chux@chgc.sh.cn